Author’s response to reviews

**Title:** Case report Peripheral Adenomatoid Odontogenic Tumor in a Cloak of an Epulis: Report of a rare case

**Authors:**

Dhara Dwivedi (dwivedidhara@gmail.com)
Nitin Prabhakar (bhopalcraniofacialinstitute@gmail.com)
Sowmya Kasetty (sowmyakasetty@gmail.com)
Rinky Ahuja (trust.rinky123@gmail.com)

**Version:** 2 **Date:** 31 Mar 2019

**Author’s response to reviews:**

RESPONSE LETTER

This is to inform you that all the grammatical and language errors have been corrected.

The additional changes recommended by the reviewers (available at the journal site) have already been made. A covering letter explaining all the changes point by point had been uploaded earlier.